RESEARCH Open Access

Stigma related to contraceptive use and abortion in Kenya: scale development and validation



Marlene Makenzius^{1*}, Grace McKinney¹, Monica Oguttu^{2,3} and Ulla Romild⁴

Abstract

Background: Stigma related to abortion and contraceptive use is a serious public health threat for young people, and validated scales to measure this stigma are scarce. The purposes of the study were to validate a newly constructed scale to measure the stigma of contraceptive use and to adapt a scale to measure the stigma of abortion.

Methods: A study nested in a cluster-randomised trial. In 2017, data was collected from 633 secondary school youths, in a semi-urban setting in western Kenya. A qualitative pre-phase (face-validity) were initially utilised to draft and validate a seven-item scale to capture contraceptive use stigma (CUS) and to adapt the Stigmatizing Attitudes, Beliefs and Actions (SABA) scale (18 items), which captures aspects of abortion stigma. Statistical tests used included test-retest reliability analysis, Pearson's correlation coefficients, Wilcoxon signed-rank test, Factor Analysis, Principal Component Analysis, interclass correlation and Cronbach's alpha.

Results: For the CUS scale, paired t-test and Wilcoxon signed-rank test showed no significant score changed between time points (p = 0.64; 0.67). CUS had similar patterns between time points, with two relevant components: *promiscuity* and *lack of autonomy*. Cronbach's alpha indicated acceptable internal consistency between time points (0.71;0.7). The confirmatory factor loadings for each item in the modified three subscales of SABA had a similar pattern to the original SABA scale, in particularly regarding negative stereotyping and, excluding and discriminating factors. The Cronbach's alpha was adequate, although lower for the modified SABA (0.74) as compared to the original SABA (0.9). The SABA scale was renamed into Adolescents Stigmatizing Attitudes, Beliefs and Action (ASABA) scale.

Conclusions: The CUS scale is considered valid and reliable for measuring contraceptive use stigma, and the ASABA scale was rated as reliable for capturing abortion stigma based on negative stereotyping and excluding and discriminating factors. The CUS, up to date the first ever proposed CUS scale, and the ASABA scale can be used to measure effects of stigma reduction interventions with the aim of preventing unintended pregnancies, motherhood and unsafe abortion among adolescents in Kenya and similar low-resource settings.

Keywords: Scale development, Validation, Social stigma, Adolescent pregnancy, Abortion, Contraception

Plain English summary

Contraceptive use stigma and abortion stigma is acknowledged worldwide, but still poorly researched. Contraceptive use stigma and abortion stigma largely nourishes from gender stereotypes, which are used to deny girls and women access to abortion and contraceptive services. In particularly the stereotype ascribing women to the role of

motherhood. This stereotype implies that women should prioritize childbearing and child caring over all other roles, even schooling. As a result (which in some cases are exacerbated by cultural and religious fundamentalism), there is a negative effect over the way a given society perceives abortion, as well as those who seek or have had an abortion, those who work in abortion care, and those who actively support abortion rights. Finally, contraceptive use stigma and abortion stigma serves to erase and disguise legitimate health services, discredit those who would provide or procure it and undermine those who advocate for

¹Department of Public Health Sciences, Global Health (IHCAR), Karolinska Institutet, Tomtebodavägen 18A, 171 77 Stockholm, Sweden Full list of author information is available at the end of the article



^{*} Correspondence: marlene.makenzius@ki.se

its legality and accessibility. The roots of such stigma that intend to control female sexuality are based on social constructs that can be deconstructed. However, as in the case of HIV and AIDS, we must empirically determine what role stigma plays and how it is expressed in particular communities before we can effectively address the problem. This paper in the issue of Reproductive Health provides validated scales measuring contraceptive use stigma (CUS) and abortion stigma (ASABA), among adolescents. Adolescents are the far most affected population suffering from contraceptive use stigma and abortion stigma. The CUS scale, up to date the first ever proposed CUS scale and the ASABA scale can be used to measure effects of stigma reduction interventions with the aim of preventing unintended pregnancies, motherhood and unsafe abortion among adolescents in Kenya and similar low-resource settings.

Background

Although access to safe abortions and effective contraceptives has increased worldwide, accessibility is still a great barrier in many low- and middle-income countries (LMIC) settings, including Kenya [1–5]. Social norms and attitudes related to contraceptive use and abortion continues to pose a serious threat to women's health and quality of life across the world [6–8]. The risk of death associated with pregnancy is about a third higher among 15- to 19-year olds as compared to women of other ages [9–11].

Adolescents in LMICs, particularly unmarried ones, face a number of barriers in obtaining contraception and using them correctly and consistently [2, 3, 9, 12, 13]. One of these barriers is a strong taboo against sexual activity outside of marriage [12, 14-16], which makes it more difficult to collect and assess unmet-need data among this group. Surveys that ask about sexual activity do not always include unmarried women, and even when they do, these women may underreport their sexual activity and contraceptive use due to an unwillingness to risk social disapproval and stigmatising attitudes associated with nonmarital sexual activities [17-19]. In addition, both women and men have concerns about the effects of contraceptive methods on women's bodies, including their weight, menstrual cycles, libido, sexual desirability and pleasure [20-23]. Such studies reveal that both men's and women's resistance to contraception could be related to traditional gender norms and power, or to a suspicion that outsiders (Westerners) aim to control women's fertility [19].

Overall, sexually active unmarried women who are not using contraception make up an estimated 18% of all women in developing countries with unmet needs for any contraception method [13], and the proportion is estimated to be even higher among sexually active unmarried

adolescents [19, 24, 25]. The majority of abortions result from unintended pregnancies (e.g., mistimed or unwanted pregnancies). Abortions occur as frequently in the mostrestrictive category of country as they do in the leastrestrictive category, 37 and 34 per 1000 women, respectively [8]. Worldwide, about 45% of all abortions are performed with unsafe procedures (e.g., by an individual who does not have the necessary training or in an environment that does not conform to minimal medical standards). Overwhelmingly, unsafe abortions occur in developing regions where restrictive abortion laws are concentrated, so as the case in Kenya [7, 8]. Abortion is not permitted in Kenya, unless, in the opinion of a trained health professional, there is need for emergency treatment, or the life or health of the mother is in danger. No safe abortion is offered in public health facility in Kenya. Safe abortion services can only be found in a few private hospitals at a higher cost and has frequently become the privilege of the rich. Abortion is still spoken to by the 1970 Penal Code which criminalises it, and the 2010 Constitution which makes exceptions to this criminalisation. The lack of clarity and transparency with regard to the circumstances in which abortion is legal greatly contributes to Kenya's high maternal mortality ratio from complications of unsafe abortion [6–10, 24]. Meeting the unmet need of modern contraception for women aged 15-19 would reduce unintended pregnancies among this age group by 6 million annually. This could prevent 2.1 million unplanned births, 3.2 million abortions (most of them unsafe) and 5600 maternal deaths [13]. Despite the fact that the age of sexual consent in Kenya is 18 years, adolescent pregnancy is one of the major causes of school dropout among girls between 16 and 18 years of age [24], as well as a major cause of death among adolescent girls [9, 10, 26]. Sexual and reproductive health services in Kenya fall short of meeting adolescents' needs. For example, an estimated 665,000 young women aged 15-19 in Kenya are married or sexually active and want to avoid becoming pregnant in the next 2 years [24]. However, increased knowledge on contraception may not be the only solution since high level of knowledge on contraceptives does not always amount to practice [27]. Contraceptive use among adolescents is sometimes associated with immorality and a promiscuous lifestyle, and the use is considered physically harmful [16, 17, 19, 20]. Such stigmatising attitudes may shame and silence young women about their contraceptive needs, which may lead to unintended pregnancies and unsafe abortions [19].

The social stigma surrounding unintended pregnancies and abortion plays a critical role in the social, medical and legal marginalisation of contraceptive services and abortion care [6, 28]. Historically, stigma theory has progressed from an individualistic focus towards the recognition of stigma as a socially constructed process [29].

Conceptualisations of sexuality and stigma are shaped at different levels within a society and are framed by a multiplicity of factors, including political structures and sociocultural context and intersections with other identities such as gender, socioeconomic status, race, ethnicity, religion and rural/urban location [30].

In 2013, Shellenberg et al. [31] developed a Stigmatizing Attitudes, Beliefs and Action (SABA) scale to measure stigmatising attitudes related to abortion. This framework referred to interrelated stigma components such as labelling, stereotyping, separating/excluding and discriminating [30]. The SABA scale was tested among adults in Ghana and Zambia [31]. The International Planned Parenthood Federation has adapted the SABA scale to the situation of young people, but it is not yet validated. To the best of our knowledge, at the beginning of this research, no validated scale or quantitative instrument specifically designed to measure CUS exists. If we can further understand and measure sociocultural stigma related to contraceptive use and abortion among the most affected population (adolescents), we will be better equipped to develop interventions to reduce such stigma.

This study aimed to develop and validate a scale to measure CUS and to validate SABAs among adolescents in western Kenya.

Methods

Study design and setting

This was a study, nested in a larger cluster randomised intervention trial (ClinicalTrial.gov NCT03065842), among secondary school youths in Kisumu, Kenya. Kisumu is the third-largest city in Kenya with a population of approximately 500,000 and bordering Lake Victoria. The overarching objective of the intervention trial was to reduce the stigma of adolescent girls using contraceptives and/or undergoing abortion procedures. For the current study the aim was to draft and validate instruments used in the trial to measure the effect of the intervention given to the students (evidence and rights based information, and value clarifications activities on abortion and contraceptive use). The project was conducted in collaboration between Karolinska Institutet in Sweden and Kisumu Medical and Education Trust (KMET), and in partnership with the Ministry of Education and the Ministry of Health in Kisumu County, located in the western region of Kenya. KMET is a nongovernmental organisation registered under the Trustee Act of 1995. KMET has no political, religious or governmental affiliation.

For the larger trial, where the current study was nested in, one intervention school and one control school were cluster-randomised from the regional sample frame of four schools. Inclusion criteria for the schools were as follows: semi-urban secondary day schools, mixed gender, with a minimum of 400 students. This study was based on data derived from the control school, which allowed data analysis without the intervention effect.

The population of Kisumu County consists primarily of the Luo and Kisii ethnic groups, mainly Christians who speak Luo, Kiswahili and English, like Kenya overall (83% Christians, 11.2% Muslims) [32]. The proportion of young people in Kisumu County is higher than the country average. That is, one in four (25%) people in Kisumu is an adolescent aged 10-19. A young population has implications for the county's health and development agenda due to the demands put on the provision of services, including health and education. In Kenya, a main area of concern is the sexual and reproductive health of adolescents and the extent to which their sexual and reproductive health and rights (SRHR) are met. Half of the women aged 20-49 and men aged 20-54 first had sex by age 16 and 18, respectively. Sexual education is an important determinant for SRHR. Girls who complete secondary school are less likely to have an unintended pregnancy and are more likely to have higher socioeconomic status. About 95% of them are enrolled in primary school, but only 61% transition to secondary school, which is about the same as the national statistics [32].

Face validity

Pre-phase work, based on a multi-method approach that was seen in the study of Shellenberg et al. [31], was conducted from November 2016 to January 2017. The procedure followed guidelines by the International Network for the Reduction of Abortion Discrimination and Stigma (Inroads) on how to adapt a scale to measure abortion stigma [33]. Participants in the pre-phase work were healthcare providers, teachers and students were invited to participate by a letter from a research assistant at KMET. Inclusion criteria's were health care staff working at youth friendly clinics (n = 30) [16], teachers and adolescents from the school catchment-area. The pre-phase work aimed to discuss participants' views on unintended pregnancy, abortion and contraceptive use among adolescents, and explore the construction of a scale to measure CUS and to explore the applicability of the SABA scale [31] in regards to adolescents in Kisumu. A total of four workshops and six focus group discussions were conducted, digitally recorded and verbally transcribed. They involved healthcare providers (n = 30), teachers (n = 19) and students (n = 21) to explore attitudes and beliefs associated with contraceptive use and abortion among adolescent girls.

Based on findings from the pre-phase work, a sevenitem CUS scale was developed—it followed a standard procedure for Likert scale development, per DeVellis [34] (five-point Liker scale: 1 = strongly disagree, 2 = disagree, 3 = unsure, 4 = agree and 5 = strongly agree). The

CUS scale captured stigmatising attitudes and beliefs regarding contraceptive use among adolescent girls. The applicability and clarity of the SABA scale was confirmed, but modified. The SABA scale captured three important dimensions of stigma related to women associated with abortion (18 items): Subscale 1, negative stereotypes (8 items); Subscale 2, discrimination and exclusion (seven items); and Subscale 3, potential contagion (three items) [31]. The format of the SABA scale was identical to the CUS scale that used a five-point Likert scale. The word woman was changed to girl in all questions, and two items were added to the scale. Further editing was done. For example, the true-or-false question (Q) A married woman is more deserving an abortion than an unmarried woman was added to Subscale 1, and Q6, A woman who has had an abortion might encourage other women to get abortions, was changed to A girl who has had an abortion might be a bad influence on other women and removed from Subscale 1 and added to Subscale 3. For Subscale 2, A girl who has had an abortion should be prohibited from going to school was added. The adapted SABA was renamed Adolescents' Stigmatizing Attitudes, Beliefs and Actions (ASABA) scale. The CUS and ASABA scales were pretested (633 adolescents) as a baseline survey, i.e. time point (TP) I and were followed up with a repeat test of 548 of the 633 adolescents after 1 month (TP II).

Statistical methods

The scales of CUS and ASABA were statistically tested. According to DeVellis [34], there are variations in suggested sample sizes needed for scale development, but a minimum of five respondents per scale item is recommended, which the current sample of 633 individuals exceeded. The reliability of the CUS scale was tested by using paired sample t-tests and a Wilcoxon signed-rank test to compare total sum scores, means (SD), and mean ranks, and for each item between time points. In addition, Kaiser-Meyer-Olkin (KMO) was performed for each survey at each of the two time points in order to verify that the data was suitable for Principle Component Analysis (PCA) [35]. Exploratory Factor Analysis (FA) with PCA is the standard procedure to validate new scales [36], and the result from TP I was compared to TP II. As there was no previous structure to follow, components with eigenvalues > 1.0 were included, and questions that had a factor loading ≥ 0.40 were retained. Oblique (promax) rotation was used, with a kappa of 4, to help create a simple structure that contained interpretable factors and to allow for correlation between the items. Cronbach's alpha was used to measure internal consistency and reliability of a psychometric [37, 38].

The statistical analysis for the ASABAs was based on the previous validation of the SABA scale in Ghana and Zambia by Shellenberg et al. [31] For that aim, a parallel-form reliability analysis was conducted to compare the results of the ASABA scale to the results of the SABA scale that was validated among heterogeneous groups from Ghana and Zambia in 2013 [31]. A confirmatory PCA analysis was used to compare responses for the total scale and for each subscale (component) [36]. The comparisons were further analysed by using Cronbach's alpha coefficients, which provided evidence of internal consistency and reliability, given that all four coefficients surpassed the minimum standard for reliability (0.60). KMO, PCA, Wilcoxon signed-rank tests and Cronbach's alpha, medians (SD) were generated with SPSS (Version 23).

Results

The cluster-randomised study comprised in total 1207 students, distributed between one intervention school (n = 574) and one control school (n = 633), both Christian secondary schools located in semi-urban areas in western Kenya. To avoid the intervention effect between time points, the analysis in the current study was based on the 633 students from the control school. Out of the 633 students, 325 (51.3%) were females, 305 (48.2%) were males, and 3 (0.5%) students did not reveal their sex. The mean age was 16.8 years (16.7 for females and 17.00 years for males).

The reliability of the CUS scale between TP I and TP II are shown in Table 1. The results shows that Q2 has a mean score of 4 at TP I but a mean score of 3.09 for TP II. In Q7, the mean score was 2.10 for TP I, compared to 1.86 for TP II. As for Questions 1, 3, 4, 5 and 6, the mean scores are stable between time points. The paired t-test (used as a test-retest measure) of the total score was not significant (mean difference of 0.13, p = 0.64), suggesting no total score-generating mechanism changes from TP I to TP II. Wilcoxon signed-rank test on total score, confirmed the paired t-test, with a p-value of 0.67.

The CUS data was found appropriate for PCA analysis, as its KMO values were 0.68 for TP I and 0.69 for TP II. As illustrated in Table 2, two components were retained under orthogonal (promax) rotation for both time points. All questions of the CUS scale have or are close to having factor loadings \geq 0.40 for both time points of one component. For Component 1, Questions 1, 2, 5 and 6 all had factor loading values \geq 0.40 at TP I and II. Questions 3 and 7 load in TP II of Component 1 and TP I of Component 2. The internal consistency for the CUS was 0.713 at TP I and 0.697 for TP II.

The SABA scale was initially adapted, and the added Q, A married woman is more deserving of an abortion than an unmarried woman had a mean of 2.04 (SD = 1.23). The Q A girl who has had an abortion should be prohibited from going to school had a mean of 1.41 (SD =

Makenzius et al. Reproductive Health (2019) 16:136 Page 5 of 10

Table 1 Mean (SD) score for each question of the CUS at TP I (n = 608) and TP II (n = 517)

Question Score 1–5		Time Point I n = 608 ^a Mean (SD)	Time Point II n = 517 ^a Mean (SD)	Wilcoxon signed- rank test
1	A girl who uses a contraceptive method is promiscuous (sexually immoral, likes to have many sexual relationships).	3.10 (1.37)	3.00 (1.33)	0.061
2	A girl who use a contraceptive method will encourage other girls to have a promiscuous lifestyle.	4.00 (1.29)	3.09 (1.31)	0.002
3	A girl cannot decide for herself to use a contraceptive method.	2.37 (1.27)	2.50 (1.29)	0.052
4	A married woman is more deserving of a contraceptive method than an unmarried woman.	2.68 (1.38)	2.63 (1.35)	0.392
5	A girl who uses contraceptives will have problems when she decides to get pregnant.	3.15 (1.45)	3.07 (1.37)	0.457
6	A girl who carries condoms is likely to have many sexual partners.	2.88 (1.48)	2.84 (1.44)	0.212
7	A girl should not insist on using a condom; it is the man to decide whether to use a condom.	(1.23)	2.10 (1.28)	< 0.001

^aOnly those who answered all questions were included in the analysis. Internal dropout rate on specific items was 1.3–2.1% (n = 8–13)

0.81), and Q6 in Subscale 1 was moved to Subscale 3. The analysis revealed that when the Q *A married woman is more deserving of an abortion than an unmarried woman* was added to Subscale 1, the KMO decreased from 0.76 to 0.74 and the factor loading was < 0.40 (0.14). Similarly, when adding the Q *A girl who has had an abortion should be prohibited from going to school* to Subscale 2, the KMO slightly decreased from 0.78 to 0.77. When Q6 was removed from Subscale 1 and added to Subscale 3, the KMO increased from 0.56 to 0.63, but the total variance explained in Subscale 3 decreased from 44.60 to 37.92. Consequently, the structure of the original SABA was retained for the ASABA scale and revealed a KMO of 0.815, with 48.96% explained variance in the data for the full 18-item scale.

Table 3 shows similar patterns between ASABA and SABA regarding the mean score of subscales and total scores. Overall, the SD for the three subscales of SABA shows a larger variation in the data compared to the ASABA scale (p < 0.001). Consequently, the scores of the ASABA scale are closer to the mean, compared to the result of the SABA scale (Table 3).

Table 2 Results of the PCA for the CUS scale at each time point with oblique (promax) rotation TP I (n = 608) and TP II (n = 517)

Question	Component 1		Component 2		
	Time Point I	Time Point II	Time Point I	Time Point II	
1	0.825	0.748	-0.206	- 0.039	
2	0.791	0.825	-0.074	-0.186	
3	-0.132	0.072	0.682	0.702	
4	0.165	0.402	0.415	0.276	
5	0.545	0.544	0.290	0.095	
6	0.434	0.434	0.398	0.288	
7	-0.093	-0.095	0.725	0.850	

Bolded factor loadings ≥0.40

Table 4 shows confirmatory factor loadings for each and one of the three subscales of ASABA in relation to the SABA scale. The three-factor structure of the ASABA scale, negative stereotyping (KMO 0.76), exclusion and discrimination (KMO 0.78) and fear of contagion (KMO 0.56), explained variance in the data with 43.34, 35.89 and 44.6%, respectively. The explained variance was 49% for the total 18-item scale (KMO 0.82). As shown in Table 4, Q1 loaded < 0.40. However, Q1 loaded stronger in a fourth component (0.9).

Table 5 presents the interclass correlations for the subscales and total scale scores for the full sample, showing significant correlations (p < 0.001-0.001).

The internal consistency coefficients for the ASABA scale compared to the SABA scale are lower for all three subscales. The ASABA subscales of negative stereotyping, exclusion and discrimination had coefficients like the ones of the SABA scale. However, the ASABA subscale of fear of contagion had a lower internal consistency coefficient than that of the SABA scale (0.38 compared to 0.80). The overall coefficient was lower for the ASABA (0.74) as compared to the SABA (0.9) (Table 6).

Discussion

The aims of this study were to develop a scale to measure CUS and to adapt and validate a scale to measure abortion stigma among adolescents. The findings strongly indicates that CUS scale is reliable to measure stigma related to contraceptive use among adolescent girls, The ASABA scale was rated reliable to capture abortion stigma based on negative stereotyping and excluding and discriminating factors. Both scales are timely relevant since the potential social, economic and health-related consequences of unintended pregnancies among adolescent girls [6, 9–11, 13] makes it essential to measure and understand unmet needs for contraception and

Table 3 Stigmatising attitudes, beliefs and actions among secondary students in Kenya (n = 633) as compared to the heterogeneous sample from Ghana and Zambia (n = 531)

Scale	items ASABA scale (SABA scale)	Kenya (n = 613) Subscale and item score Mean (SD)	Ghana and Zambia ^a (n = 531) Subscale score Mean (SD)
Stigm	atising attitudes, 8 items (min-max = 8-40)	28.03 (5.76)	25.7 (7.48)
1	A girl who has an induced abortion is committing a sin.	4.49 (1.06)	
	(A woman who has an abortion is committing a sin.)		
2	Once a girl has one abortion, she will make it a habit.	3.38 (1.31)	
	(Once a woman has one abortion, she will make it a habit.)		
3	A girl who has an abortion cannot be trusted.	3.05 (1.42)	
	(A woman who has had an abortion cannot be trusted.)		
4	A girl who has an abortion brings shame to her family.	3.92 (1.23)	
	(A woman who has had an abortion brings shame to her family.)		
5	The health of a girl who has an abortion is never as good as it was before the abortion.	3.78 (1.36)	
	(The health of a woman who has had an abortion is never as good as it was before the abortion.)		
6	A girl who has had an abortion might be a bad influence on other women.	3.29 (1.41)	
	(A woman who has had an abortion might encourage other women to get abortions.)		
7	A girl who has an abortion will be a bad mother.	2.62 (1.36)	
	(A woman who has an abortion is a bad mother.)		
8	A girl who has an abortion brings shame to her community.	3.46 (1.36)	
	(A woman who has an abortion brings shame to her community.)		
exclus	sion and discrimination, 8 items (min-max = 8–40)	14.31 (4.66)	15.72 (5.79)
9	A girl who has had an abortion should be prohibited from going to religious services. (A woman who has had an abortion should be prohibited from going to religious services.)	1.54 (0.88)	
	A girl who has had an abortion should be teased so that she will be ashamed about her decision.	2.14 (1.29)	
10	(I would tease a woman who has had an abortion so that she will be ashamed about her decision.)		
	A girl should be disgraced in my community if she has had an abortion.	2.09 (1.17)	
11	(I would try to disgrace a woman in my community if I found out she had an abortion.)		
	A man should not marry a woman who has had an abortion.	2.06 (1.16)	
12	(A man should not marry a woman who has had an abortion because she may not be able to bear children.)		
	A girl who has had an abortion should no longer be associated with me.	1.90 (1.12)	
13	(I would stop being friends with someone if I found out that she had an abortion.)		
14	A girl who had an abortion should be pointed fingers at so that other people would know what she has done.	1.81 (1.12)	
	(I would point my fingers at a woman who had an abortion so that other people would know what she has done.)		
	A girl who has an abortion should not be treated the same as everyone else.	1.75 (1.04)	
15	(A woman who has an abortion should not be treated the same as everyone else.)		
ear o	of contagion, 3 items (min-max = 3–15)	5.86 (2.19)	7.54 (3.41)
	A girl who has had an abortion can make other people fall ill or get sick.	1.68 (1.04)	
16	(A woman who has an abortion can make other people fall ill or get sick.)		
17	A girl who has had an abortion should be isolated from other people in the community for at least 4 weeks after having an abortion.	2.20 (1.12)	
	(A woman who has an abortion should be isolated from other people in the community for at least 1 month after having an abortion.)		
	If a boy has sex with a girl who has had an abortion, he will most likely become infected with a disease.	1.96 (1.11)	

Table 3 Stigmatising attitudes, beliefs and actions among secondary students in Kenya (n = 633) as compared to the heterogeneous sample from Ghana and Zambia (n = 531) (Continued)

Scale items ASABA scale (SABA scale)	Kenya	Ghana and
	(n = 613)	Zambia ^a
	Subscale and	(n = 531)
	item score	Subscale score
	Mean (SD)	Mean (SD)
18 (If a man has sex with a woman who has had an abortion, he will become infected with a disease.)		
Total scores (min-max = 18–85)	47.19 (9.27)	48.9 (14.2)

^aInternal dropout on specific items 0.0–1.7%

safe abortions. Although the Kenyan Constitution offers potential for increasing women's access to safe abortion, societal attitudes towards young women who procure abortion remain largely negative. While stigma, and access barriers to safe abortion and modern contraceptives continues to be neglected, young women will not only delay the request for termination but also use resort to unsafe and life threaten options.

The two components of the CUS were labelled *promiscuity* and *lack of autonomy*. These labels were chosen based on the themes shared by the questions contained in each component. Most of the questions that loaded in Component 1 share the theme of contraceptive use being associated with promiscuity, which can be seen

Table 4 Results of confirmatory analysis of the ASABAs (n = 613), analysed with principal component analysis with direct oblique rotation, among secondary students in Kenya (n = 633), compared to the heterogeneous sample from Ghana and Zambia (n = 531)

Question	Subscale 1		Subscale 2		Subscale 3	
	ASABA	SABA	ASABA	SABA	ASABA	SABA
1	.202ª	.750	-	_	-	-
2	.448	.630	-	-	-	-
3	.669	.720	-	_	-	-
4	.680	.480	-	-	-	-
5	.539	.610	-	_	-	-
6	.452	.460	-	_	-	-
7	.628	.740	-	-	-	-
8	.631	.670	-	-	-	-
9	-	-	.467	.610	-	-
10	-	-	.560	.660	-	-
11	-	_	.663	.530	-	_
12	-	-	.601	.820	-	_
13	-	-	.646	.620	-	-
14	-	_	.676	.580	-	_
15	-	-	.553	.640	-	-
16	-	-	-	-	.600	.870
17	_	-	-	-	.656	.590
18	_	-	-	-	.740	.870

^aLoaded 0.9 in fourth component

either in the characteristics of girls who use contraceptives or in the belief that girls who use contraceptives encourage others to be promiscuous. However, Q5, A girl who uses contraceptives will have problems when she decides to get pregnant, does not fit this label as well as the other questions. At face value, this question may not relate to promiscuity, as it proposes that contraceptive use will result in fertility issues later in life. However, the issue can be looked at from another angle. A girl who uses a contraceptive method is promiscuous claims that a girl who uses contraceptives is inherently promiscuous. Taking this into account, if girls who use contraceptives are promiscuous and are therefore at greater risk for catching STIs, this would impact their ability to have children later in life [39].

Component 2 was labelled *lack of autonomy*, suggesting that a girl is not supposed to, or should not be allowed to, make the decision to use contraception, and that this responsibility should be put on someone else (for example, her partner or her parents). The distribution of power in the developing world (and particularly in sub-Saharan Africa) is unequal, and men are often the primary decision makers about family size and the use of family planning.

Some differences in the results between the scales of ASABA and SABA may be explained by different sociodemographic samples. The original SABA scale was tested on a heterogeneous sample, while the ASABA scale was tested on a homogenous sample. The SABA scale was tested in Ghana and Zambia among participants with different background characteristics [31], whereas the sample that tested the ASABA scale came only from the Kisumu region of Kenya. Specifically, the respondents for the ASABA scale all came from a Christian secondary school. Moreover, the age range for the sample of SABA scale was larger, and the original sample included individuals aged 18 to 54 [31], while the sample of the current ASABA scale included students aged 13 to 21. People of different ages tend to have different views about social issues because of where they are in their lives and their level of openness [40]. Also, older generations tend to be more conservative regarding social issues as compared to younger generations [41]. The larger variety in age, as well as the higher

Makenzius et al. Reproductive Health (2019) 16:136 Page 8 of 10

Table 5 Interclass correlations for the three subscales and the full ASABA scale

Subscales	Negative stereotypes (n = 613)	Exclusion and discrimination (<i>n</i> = 603)	Fear of contagion (<i>n</i> = 612)	Total ASABA (n = 602)
Negative stereotypes	1	.231 < .001 ^a	.138.001 ^a	.772 < .001 ^a
Exclusion and discrimination	.231 < .001 ^a	1	.477 < .001 ^a	.764 < .001 ^a
Fear of contagion	.138.001 ^a	.477 < .001 ^a	1	.566 < .001 ^a
Total ASABA	.772 < .001 ^a	.764 < .001 ^a	.566 < .001 ^a	1 < .001 ^a

^a Correlation significant at .05 level

mean age, in the SABA sample could therefore have yielded different results than the younger ASABA sample.

Overall, the similarities in loadings in SABA and ASABA were meant to retain the three components (subscales) from the original SABA on the ASABA. These were negative stereotyping, exclusion and discrimination and fear of contagion. The adapted version did not seem to add relevant information that could strengthen the tool. Most items on ASABA loaded similar to what they loaded on the SABA, despite their adaption to a new population, with one exception. First, the item in Subscale 1: A girl who has had an abortion is committing a sin, seems to have an overarching position, as it loaded 0.9 in the fourth component. Most students held the opinion that abortion is committing a sin, regardless if their overall opinions were not stigmatising. Most people in Kenya are highly religious, and teaching in schools is faith-based. The applicable school was a Christian school, and its teaching is based on Christian principles and values, which may explain this item's overarching role in the scale. Given this, the underlying themes and the subscales they define are considered to be the same as the SABA scale. However, the use of ASABA should provide the option to exclude the third subscale, fear of contagion, since this subscale had a notable lower internal consistency compared to the result found in the sample of Ghana and Zambia [31].

One limitation in comparing the ASABA to the SABA is the unclear validation procedure of Shellenberg et al. [31] Shellenberg et al. explains that the data from one time point was explored by using an exploratory FA and then a PCA by using promax rotation, but the description of how

Table 6 Internal consistency coefficients for ASABA, compared to SABA (n = 613)

Subscales	Set of items	Cronbach's alpha	
	ASABA (SABA)	ASABA (SABA)	
Negative stereotyping of women	8 (8)	0.67 (0.85)	
Exclusion and discrimination	7 (7)	0.70 (0.8)	
Fear of contagion	3 (3)	0.38 (0.8)	
Total items	18 (18)	0.74 (0.9)	

components were retained is vague [31]. Part of how they decided to retain components was by way of 'interpretability factors' [31]. These factors were likely judgement calls that cannot be replicated. For this reason, parallel analysis was used for comparison and may not have been analysed in a comparable way to the original SABA validation study. This, however, is clearly not a direct limitation of our data. It simply limited our ability to fairly compare with the results of Shellenberg et al. Additionally, although the test-retest is a strength, the follow-up time of 1 month could be considered too long. The length of time between the first testing and the retest is critical [41]. A long follow-up time allows for true changes in attitude [41]. Secondary students may also be impressionable, and their views change frequently as they gain life experience (including gaining an education) or interactions with family and friends. This could have influenced the measure of the stability of the scales. A final limitation was the presence of missing data, which may account for some of the differences between time points for CUS. For both time points, not all the participants answered all the questions. Although the missing values were considered to be by random chance and not systemic in nature, participants with missing data were excluded in the paired sample analysis to minimise their effect. As a result, data for some individuals was not used in the analysis, which caused the sample size to be less than the total number of participants and which could have had an impact on the findings.

The CUS scale and the ASABA scale may be useful to measure societal attitudes and cultural beliefs around gender roles related to SRHR. The information can be used in the design of behaviour change campaigns and comprehensive sexual education, in the development and introduction of new contraceptive methods, and in provider training and service delivery, including counselling about methods. In the long run, this may increase female empowerment at community and household levels for a sustainable equality movement regarding SRHR.

Conclusion

The CUS scale is considered validated and reliable to measure contraceptive use, based on attitudes that suggest that girls who use contraceptives are potentially promiscuous and that girls lack autonomy to decide whether to use contraceptives. The ASABA scale is rated reliable to capture abortion stigma based on negative stereotyping and excluding and discriminating factors. However, further validations may be needed to investigate the potential impact that fear of contagion may have on abortion stigma. The CUS and ASABA scales can be used to measure effects of stigma reduction interventions with the aim to prevent unintended pregnancy, motherhood and unsafe abortion among adolescents in Kenya and similar low-resource settings.

Acknowledgements

The article is partly based on Grace McKinney's master thesis, Department of Public Health Sciences, spring term 2018 at Karolinska institutet, Sweden. The authors wish to express their sincere appreciation to the staff of Kisumu Medical Education Trust (KMET) for their assistance during data collection in Kisumu, and to all students who participated in the study.

Authors' contributions

MM: Conception and design of the work, and substantial contributed to analysis, interpretation of data for the work, and drafting the work. GMcK: Contributed substantially to analysis, interpretation of data for the work, and drafting the work. MO: Critically revised for important intellectual content. UR: Advised and supervised on statistical analysis, and critically revised for important intellectual content. All authors above approved to this final version to be published, and agreed to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.

Funding

This study was funded by the Swedish Research Council for Health, Working Life and Welfare (2015-01194), and the Swedish Research Council (2016-05670).

Availability of data and materials

Data is yet to be analysed for other publications, hence the availability of data is currently limited. Contact marlene.makenzius@ki.se to request data availability.

Ethics approval and consent to participate

To ensure the safety and autonomy of all participants, everyone was asked to give informed consent to participate. Given that the study population comprised adolescents, participants under the age of 18 were required to provide a signed assent and their tutors' consent in order to participate. The informed consent document was translated into Kiswahili and then back to English to provide a clear understanding for all participants, as well as an accurate translation. While it was not possible to completely anonymise the data, the research team ensured that all data was stored securely and according to Kenyan laws and guidelines. Ethical approval was given by the Jaramogi Oginga Odinga Teaching and Referral Hospital Ethical Review Committee (ERC.1B/VOL.1/263) and the Kenyan National Commission for Science Technology and Innovation.

Competing interests

The authors declare that they have no competing interests.

Author details

Department of Public Health Sciences, Global Health (IHCAR), Karolinska Institutet, Tomtebodavägen 18A, 171 77 Stockholm, Sweden. ²College of Health Sciences, School of Nursing Sciences, University of Nairobi, Nairobi, Kenya. ³Kisumu Medical Education Trust – KMET, Kisumu, Kenya. ⁴The Public Health Agency of Sweden, Ostersund, Sweden.

Received: 4 April 2019 Accepted: 27 August 2019 Published online: 06 September 2019

References

- Cleland J, Conde-Agudelo A, Peterson H, Ross J, Tsui A. Contraception and health. Lancet. 2012;380(9837):149–56.
- Chandra-Mouli V, McCarraher DR, Phillips SJ, et al. Contraception for adolescents in low- and middle-income countries: needs, barriers, and access. Reprod Health. 2014;11(1):1.
- Sedgh G, Hussain R. Reasons for contraceptive nonuse among women having unmet need for contraception in developing countries. Stud Fam Plan. 2014;45(2):151–69.
- Ganatra B. Health worker roles in safe abortion care and post-abortion contraception. Lancet Glob Health. 2015;3(9):e512–3.
- Heller R, Cameron S, Briggs R, et al. Postpartum contraception: a missed opportunity to prevent unintended pregnancy and short inter-pregnancy intervals. J Fam Plann Reprod Health Care. 2016;42(2):93–8.
- Grimes DA, Benson J, Singh S, et al. Unsafe abortion: the preventable pandemic. Lancet. 2006;368(9550):1908–19.
- Sedgh G, Singh S, Shah IH, et al. Induced abortion: incidence and trends worldwide from 1995 to 2008. Lancet. 2012;379(9816):625–32.
- Sedgh G, Bearak J, Singh S, et al. Abortion incidence between 1990 and 2014: global, regional, and subregional levels and trends. Lancet. 2016; 388(10041):258–67.
- Nove A, Matthews Z, Neal S, Camacho AV. Maternal mortality in adolescents compared with women of other ages: evidence from 144 countries. Lancet Glob Health. 2014;2(3):e155–64.
- World Health Organization, 'Mortality, morbidity and disability in adolescence', (2014); http://apps.who.int/adolescent/second-decade/ section3/page2/mortality.html.
- Say L, Chou D, Gemmill A, et al. Global causes of maternal death: a WHO systematic analysis. Lancet Glob Health. 2014;2(6):e323–33.
- Yegon EK, Kabanya PM, Echoka E, et al. Understanding abortion-related stigma and incidence of unsafe abortion: experiences from community members in Machakos and Trans Nzoia counties Kenya. Pan Afr Med J. 2016;24:258
- Darroch JE, Woog V, Bankole A, et al. 'Adding it up: costs and benefits of meeting the contraceptive needs of adolescents', Guttmacher Institute (2016); available 2019-02-01 at https://www.guttmacher.org/report/addingit-meeting-contraceptive-needs-of-adolescents .
- Osur J, Baird TL, Levandowski BA, et al. Implementation of misoprostol for postabortion care in Kenya and Uganda: a qualitative evaluation. Glob Health Action. 2013;6:1–11.
- Izugbara C, Egesa C, Okelo R. High profile health facilities can add to your trouble: women, stigma and unsafe abortion in Kenya. Soc Sci Med. 2015; 141:9–18.
- Hakansson M, Oguttu M, Gemzell-Danielsson K, Makenzius M. Human rights versus societal norms: a mixed methods study among healthcare providers on social stigma related to adolescent abortion and contraceptive use in Kisumu, Kenya. BMJ Glob Health. 2018;3(2):e000608.
- 17. Cleland J, Harbison S, Shah IH. Unmet need for contraception: issues and challenges. Stud Fam Plan. 2014;45(2):105–22.
- Beguy D, et al. Inconsistencies in self-reporting of sexual activity among young people in Nairobi, Kenya. J Adolesc Health. 2009;45(6):595–601.
- Hussain R, et al. 'Unmet need for contraception in developing countries: examining women's reasons for not using a method', Guttmacher Institute (2016); available 2019-02-01 at http://www.guttmacher.org/report/unmet-need-for-contraception-in-developing-countries.
- Wells E. Countering myths and misperceptions about contraception.
 Outlook Reprod Health. 2015; available 2019-02-01 at http://www.path.org/publications/files/RH_outlook_myths_mis_june_2015.pdf.
- John NA, Babalola S, Chipeta E. Sexual pleasure, partner dynamics and contraceptive use in Malawi. Int Perspect Sex Reprod Health. 2015;41(2):99–107.
- National Council for Population and Development. 2014 Survey on Male Involvement in Family Planning and Reproductive Health in Kenya. Nairobi; 2017. Availible 2019-09-02 at http://www.ncpd.go.ke/wp-content/uploads/201 7/09/2014-National-Survey-on-Male-Involvement-in-FP-and-RH-in-Kenya.pdf.
- Kabagenyi A, et al. Barriers to male involvement in contraceptive uptake and reproductive health services: a qualitative study of men and women's perceptions in two rural districts in Uganda. Reprod Health. 2014;11(1):21.

- Statistics KNBo, Kenya Demographic and Health Survey 2014, Kenya National Bureau of Statistics (2014); available 2019-02-27 at https:// dhsprogram.com/pubs/pdf/fr308/fr.pdf.
- Makenzius M, Oguttu M, Klingberg-Allvin M, et al. Post-abortion care with misoprostol—equally effective, safe and accepted when administered by midwives compared to physicians: a randomised controlled equivalence trial in a low-resource setting in Kenya. BMJ Open. 2017;7:e016157. https:// doi.org/10.1136/bmjopen-2017-016157.
- Ziraba AK, Izugbara C, Levandowski BA, et al. Unsafe abortion in Kenya: a cross-sectional study of abortion complication severity and associated factors. BMC Pregnancy Childbirth. 2015;15:34.
- Murigi M, Butto D, Barasa S, Maina E, Munyalo B. Overcoming barriers to contraceptive uptake among adolescents: the case of Kiambu County, Kenya. J Biosci Med. 2016;4:1–10. https://doi.org/10.4236/jbm.2016.49001.
- 28. Yang LH, Kleinman A, Link BG, et al. Culture and stigma: adding moral experience to stigma theory. Soc Sci Med. 2007;64(7):1524–35.
- Link BG, Yang LH, Phelan JC, et al. Measuring mental illness stigma. Schizophr Bull. 2004;30(3):511–41.
- 30. Kumar A, Hessini L, Mitchell EM. Conceptualising abortion stigma', *Culture*. Health Sex. 2009;11(6):625–39.
- Shellenberg KM, Hessini L, Levandowski BA. Developing a scale to measure stigmatizing attitudes and beliefs about women who have abortions: results from Ghana and Zambia. Womens Health. 2014;54(7):599–616.
- Fact Sheet. Adolescent sexual and reproductive health in Kisumu. Available 2019-02-01 at https://www.afidep.org/download/Fact-sheet_Kisumu-CountyF.pdf.
- Inroads. So you want to adapt a scale to measure abortion stigma? Research & Tools. The International Network for the Reduction of Abortion Discrimination and Stigma. (2016). Available 2019-02-27 at https://endabortionstigma.org/en/Making-inroads/2016/August/scale-adaptation.aspx.
- 34. DeVellis R. Scale development: theory and applications. 3rd. Thousand Oaks: Sage Publications; 2012. [Google Scholar]
- Kaiser MO. Kaiser–Meyer–Olkin measure for identity correlation matrix. J R Stat Soc. 1974:52:296–8.
- Beavers AS, Lounsbury JW, Richards JK, et al. Practical considerations for using exploratory factor analysis in educational research. Pract Assess Res Eval. 2013;18.
- Bland JM, Altman DG. Measuring agreement in method comparison studies. Stat Methods Med Res. 1999;2:135–60.
- Tavakol M, Dennick R. Making sense of Cronbach's alpha. Int J Med Educ. 2011;2:53–5.
- Manfredi R, Beltrami C, D'Antuno A, et al. Sexually transmitted diseases (STD) and their relationship with sexual behaviour and condom use, in a cohort of teenagers referring to a STD Centre: a nine-year, prospective study. Le infezioni Med. 2001;9(3):147–53.
- Cornelis I, Van Hiel A, Roets A, Kossowska M. Age differences in conservatism: evidence on the mediating effects of personality and cognitive style. J Personality. 2009;77:51–88.
- 41. Roberts BW, Walton KE, Viechtbauer W. Patterns of mean-level change in personality traits across the life course: a meta-analysis of longitudinal studies. Psychol Bull. 2000;132(1):132.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

